Title: A lactate shuttle system between tumour and stromal cells is associated with poor prognosis in prostate cancer

Authors:

Nelma Pértega-Gomes (nelmagom@gmail.com)
José R Vizcaino (jrvizcaino@gmail.com)
Jan Attig (jattig@mtc-lmb.cam.ac.uk)
Sarah Jurmeister (sarah.jurmeister@cruk.cam.ac.uk)
Carlos Lopes (calopes@icbas.up.pt)
Fátima Baltazar (fbaltazar@ecsaude.uminho.pt)

Version: 5 Date: 6 May 2014

Author's response to reviews: see over
Dear Editor,

We are grateful for the opportunity to resubmit our Manuscript entitled “A lactate shuttle system between tumour and stromal cells is associated with poor prognosis in prostate cancer”, Ms. 2369854151138229. Our response to the editorial requirements is attached below. We look forward to hearing from you again soon with your final decision.

Yours sincerely,

Fátima Baltazar

Assistant Professor, PhD
Life and Health Sciences Research Institute (ICVS), School of Health Sciences, University of Minho
4710-057 Braga
Portugal

Editorial Requirements:

Thank you for your revised manuscript, however we have found further grammatical errors, in particular the results section of your abstract. We strongly recommend a further copy-edit.

R: The manuscript has been again copy-edited and the text was thoroughly reviewed. We hope this time the language meets the required English standard.